Author's response to reviews

Title: ARG098, a novel anti-human Fas antibody, suppresses synovial hyperplasia and prevents cartilage destruction in a severe combined immunodeficient-HuRAg mouse model

Authors:

Noriko Odani-Kawabata (noriko.odani@santen.co.jp)
Miwa Takai-Imamura (miwa.takai@santen.co.jp)
Osamu Katsuta (osamu.katsuta@santen.co.jp)
Hiroshi Nakamura (nakamura@nms.ac.jp)
Kusuki Nishioka (toranomon-nishioka@cpost.plala.or.jp)
Keiko Funahashi (tiken@mayflower-hp.jp)
Tsukasa Matsubara (mats@d2.dion.ne.jp)
Minoru Sasano (sasanom@santen.co.jp)
Hiroyuki Aono (aonoh@santen.co.jp)

Version: 6 Date: 25 September 2010

Author's response to reviews: see over
September 25, 2010

Professor John Isaacs,
Ms Judith Gorton,
The BioMed Central Editorial Team,
The BMC Series Editorial Production Team

Dear Professor Isaacs,
Dear Ms Gorton,
Dear the BMC Series Editorial Production Team,

RE: MS: 4752772533600702 entitled "ARG098, a novel anti-human Fas antibody, suppresses synovial hyperplasia and prevents cartilage destruction in a severe combined immunodeficient-HuRAg mouse model"

Thank you for acceptance of the above manuscript in principle for BMC Musculoskeletal Disorders. I am very pleased to hear it.
We are attaching a revised manuscript for the publication.
Thank you for your supports.

Sincerely yours,

Noriko Odani-Kawabata
Research and Development Center
Santen Pharmaceutical Co. Ltd.
8916-16 Takayama-cho, Ikoma-shi, 630-0101 Nara, Japan
Tel: +81-743-79-4595, Fax: +81-743-79-4507
E-mail: noriko.odani@santen.co.jp